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All Production Running
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SeaQuest Integrated Protons

Protons seen by G2SEM: 8.7E+17
. NOt Inhibited: 5.5E+17 (63%)
. and not Busy: 4.2E+17 (48%)
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Previous Week

SeaQuest Integrated Protons
1/26/15 - 2/1/15

Protons seen by G2SEM: 4.6E+16
... hot Inhibited: 2.7E+16 (59%)
... and not Busy: 2.0E+16 (45%)




Hydrogen Target

»  Sunday Morning - Pressure spiked and H2 gas vented
- LH2 target left not completely full, so LH2 was removed from target rotation
- Data continued to be recorded on other targets
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Hydrogen Target

»  Sunday Morning - Pressure spiked and H2 gas vented
LH2 target left not completely full, so LH2 was removed from target rotation
Data continued to be recorded on other targets

»  Sunday Afternoon - Attempt to fill-up LH2 failed due to obstruction in supply line

Air got into line and froze
»  Monday-Tuesday - Warmed up LH2 target to melt and eliminate blockage
»  Wednesday - Hunted for and found leak: a faulty needle valve
»  Thursday - Successfully replaced the faulty valve. Begin fill-up and cool down of LH2
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Hydrogen Target

»  Sunday Morning - Pressure spiked and H2 gas vented
LH2 target left not completely full, so LH2 was removed from target rotation
Data continued to be recorded on other targets

»  Sunday Afternoon - Attempt to fill-up LH2 failed due to obstruction in supply line
Air got into line and froze

»  Monday-Tuesday - Warmed up LH2 target to melt and eliminate blockage
»  Wednesday - Hunted for and found leak: a faulty needle valve
»  Thursday - Successfully replaced the faulty valve. Begin fill-up and cool down of LH2

» Friday Morning - Resume taking data with LH2 target

»  Thank you to AD rad safety (esp. Dale White) for 3 accesses!
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Other News

»  Muffin fan on H2 diffusion pump was replaced
»  Magnets tripped twice within an hour due to water cooling issues

» One new Linux VME in use on detector with no issues - very good news
Recall our need to migrate from VxWorks to Linux for future VMEs

»  All subsystems working well
Target, DAQ, Hodoscopes, Drift Chamber, nearonline software

» Beam has been stable
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